Response to Reviewers

Reviewer#1:
Specific Comments to Authors: The article is about challenges faced by families of health care workers including nurses during the pandemic. This is a truth that all across the world the healthcare professionals (from bottom to top) worked tirelessly and their efforts must be appreciated. In my view, this is an appraisal or gratitude towards family members of the health professionals. So, title may be modified to give the letter more clarity. Rest of the contents are good.

Thank you. The Title will be changed to “Impact of the COVID-19 Pandemic on Healthcare Workers Families”.

Reviewer#2:
Specific Comments to Authors: The findings of this review are novel. This is a topic to which very little attention has been paid and I recommend this article for publication. The literature review was well done and the data adequately synthesized.

Thank you.

Reviewer#3:
Specific Comments to Authors: Interesting and well-written.

Thank you.
Response to Editor-in-chief:

Company editor-in-chief:

I have reviewed the Peer-Review Report, full text of the manuscript, and the relevant ethics documents, all of which have met the basic publishing requirements of the World Journal of Clinical Cases, and the manuscript is conditionally accepted. I have sent the manuscript to the author(s) for its revision according to the Peer-Review Report, Editorial Office’s comments and the Criteria for Manuscript Revision by Authors. Before final acceptance, when revising the manuscript, the author must supplement and improve the highlights of the latest cutting-edge research results, thereby further improving the content of the manuscript. To this end, authors are advised to apply a new tool, the RCA. RCA is an artificial intelligence technology-based open multidisciplinary citation analysis database. In it, upon obtaining search results from the keywords entered by the author, "Impact Index Per Article" under "Ranked by" should be selected to find the latest highlight articles, which can then be used to further improve an article under preparation/peer-review/revision. Please visit our RCA database for more information at: https://www.referencerecitationanalysis.com/

Thank you. A search was done in the above database. Additional references were found and added.


